Author's response to reviews

Title: Evaluation of in vitro and in vivo Biological Activities of Cheilanthes albomarginata Clarke

Authors:

- Ramakanta Lamichhane (clickrama@hotmail.com)
- Se-Gun Kim (kimsegun@gmail.com)
- Amrit Poudel (amritpoudel@yahoo.com)
- Dipak Sharma (sharma.dipak69@gmail.com)
- Kyung Hee Lee (fuhaha112@naver.com)
- Hyun-Ju Jung (hyun104@wku.ac.kr)

Version: 5 Date: 5 March 2014

Author's response to reviews:

the ethical statement written in the paragraph - Animal and experiment design has been changed as follow:-

The whole procedure of animal experiments was authorized by Wonkwang Animal Care and Use Committee and carried out according to the "Guide for Care and Use of Laboratory Animals" published by the National Institutes of Health.